Author's response to reviews

Title: Decompressive cervical Laminectomy and Lateral Mass Screw-rod arthrodesis. Surgical Analysis and outcome.

Authors:

Moh’d M Al barbarawi (dr_barbarawi@yahoo.com)
Ziad A Audat (ziadaudat2000@yahoo.com)
Moatasem M Obaidat (moatasem@yahoo.com)
Tareq M Qudsieh (dr_tareqqod@yahoo.com)
Waleed F Dabbas (waleeddabbas@yahoo.ca)
Moanes H Obaidat (moaneso@yahoo.com)
Anas A Malkawi (asmalkawi1984@yahoo.com)

Version: 4 Date: 1 May 2011

Author's response to reviews:

the scoliosis editorial team,

dear editors,

thanks again for accepting this article for publication i went through the comments and amended the manuscript as follows,

1-the color mark was taken off.
2- names corrected and affiliation institution and departments of authors redited
3- corresponding author in a separate page was re added.
4-declaration title was replaced with competinf interests title.
5- figure remarks were removed and reedited.
6- tables added to the manuscripts and legends added below each one.

thanksa gain

yours,

dr barbarawi